Author's response to reviews

Title: Association of HLA-B*5801 Allele and Allopurinol-Induced Stevens Johnson Syndrome and Toxic Epidermal Necrolysis: A Systematic Review and Meta-analysis

Authors:

Ratchadaporn Somkrua (racha_rx@hotmail.com)
Elizabeth E. Eickman (elizapoo@hotmail.com)
Surasak Saokaew (saokaew@gmail.com)
Manupat Lohitnavy (manupatl@gmail.com)
Nathorn Chaiyakunapruk (nui@u.washington.edu)

Version: 3 Date: 1 September 2011

Author's response to reviews: see over
September 2, 2011

Author list: Ratchadaporn Somkrua, Elizabeth E. Eickman, Surasak Saokaew, Manupat Lohitnavy and Nathorn Chaiyakunapruk

Title: Association of HLA-B*5801 Allele and Allopurinol -Induced Stevens Johnson Syndrome and Toxic Epidermal Necrolysis: A Systematic Review and Meta-analysis

Journal: BMC Medical Genetics
MS ID: 1741206997521763

Dear editor:

We are writing to inform you that we have made changes of the format of our manuscript in accordance with your major revision request.

1) We have reformatted our abstract to use these headings: Background, Methods, Results, and Conclusions.

2) We inserted authors’ contribution as suggested by the editor.

3) We removed shading and colours within the tables and make it bold instead.

4) We removed the additional file (PRISMA Statement Checklist) from the system

5) We have removed all highlighting and tracking from the manuscript.

Regarding other request for minor revision, we will leave these to the editorial team to make changes as needed.

Should you have any additional changes you want us to revise, please let us know. I look forward to seeing the final edited manuscript before final dissemination.

Regards,

Nathorn